Author’s response to reviews

Title: A study on spatiotemporal delay of hand foot and mouth disease in response to weather variation based on SVD: a case study in Shandong province, China

Authors:

Yilan Liao (liaoyl@lreis.ac.cn)
Renbin Ouyang (renbinouyang@126.com)
Jinfeng Wang (wangjf@lreis.ac.cn)
Bing Xu (xub@lreis.ac.cn)

Version: 6 Date: 19 January 2015

Author’s response to reviews: see over
Response to reviewer’s comments

Q: Please change the statement in the manuscript and say "CDC China granted us permission to use the HFMD incidence data".
A: we revised this sentence ‘The HFMD incidence data came from China CDC and they promised us to use those data in the study.’ To “CDC China granted us permission to use the HFMD incidence data” in the part ‘Hand, foot and mouth disease data and meteorological data’ of the section ‘Methods’.